Author's response to reviews

Title: Comparison of dynamic navicular motion measured using the stretch-sensor between barefoot, shod, treadmill and over-ground walking and running: A cross sectional study

Authors:

Christian J Barton (christian@completesportscare.com.au)
Simon L Kappel (iha@simonk.dk)
Peter Ahrendt (pah@eng.au.dk)
Ole Simonsen (ohs@rn.dk)
Michael S Rathleff (michaelrathleff@gmail.com)

Version: 2  Date: 4 August 2014

Author's response to reviews:

The following sentence has been added to the Methods section of the manuscript: "The study was approved by Aarhus University and conducted in accordance with the Helsinki Declaration and all participants were given written and verbal information about the project and signed an informed consent before participating."